Author’s response to reviews

Title: A protein interaction map identifies existing drugs targeting SARS-CoV-2

Authors:
claudia cava (claudia.cava@ibfm.cnr.it)
Gloria Bertoli (gloria.bertoli@ibfm.cnr.it)
Isabella Castiglioni (isabella.castiglioni@ibfm.cnr.it)

Version: 3 Date: 08 Aug 2020

Author’s response to reviews:

Editor Comments:
Please proofread the article to be published, make especial attention in edition mistakes, and also add to the drug reported as a possible antiviral compounds, a detailed drug status in the drug development process, like if is an approved drug, by FDA for the use in….

We thank the editor for the suggestion. We proofreaded the article. Furthermore, we added some information such as FDA and clinical trials for the drugs reported in the Section “Discussion” lines 308-309, lines 318-321, lines 329-331, lines 342-344.